Author's response to reviews

Title: Test characteristics and potential impact of the urine LAM lateral flow assay in HIV-infected outpatients under investigation for TB and able to self-expectorate sputum for diagnostic testing

Authors:

Jonathan Peter (Jonny.Peter@uct.ac.za)
Grant Theron (Grant.Theron@uct.ac.za)
Duncan Chanda (duncanchanda@gmail.com)
Petra Clowes (pclowes@nimr-mmrc.org)
Andrea Rachow (arachow@mmrp.org)
Maia Lesosky (lesosky@gmail.com)
Michael Hoelscher (hoelscher@lrz.uni-muenchen.de)
Peter Mwaba (pbmwaba2000@yahoo.com)
Alex Pym (alexanderpym@hotmail.com)
Keertan Dheda (keertan.dheda@uct.ac.za)

Version: 6 Date: 5 December 2014

Author's response to reviews:

This manuscript has not yet been reviewed by BMC Medicine or infectious diseases. It has been referred from BMC medicine to BMC infectious diseases without peer-review